Author's response to reviews

Title: Effect of standing posture during whole body vibration training on muscle morphology and function in older adults: a randomised controlled trial

Authors:

Monica Mikhael (mmik4082@uni.sydney.edu.au)
Rhonda Orr (rhonda.orr@sydney.edu.au)
Fleur Amsen (fleuramsen@gmail.com)
David Greene (david.greene@acu.edu.au)
Maria A Fiatarone Singh (maria.singh@sydney.edu.au)

Version: 4 Date: 12 October 2010

Author's response to reviews: see over
Dear Editor

Re: MS: 5333082713385789

Research article

Effect of standing posture during whole body vibration training on muscle morphology and function in older adults: a randomised controlled trial.

Monica Mikhael, Rhonda Orr, Fleur Amsen, David Greene and Maria A Fiatarone Singh

The following revisions were made to comply with the journal’s requirements:

Major revisions
1. A Conclusion heading was inserted before the last paragraph
2. Tables 1-4 have been added to the manuscript

Minor revisions
3. Track changes have been removed
4. Affiliations have been written in full with complete postal address
5. Visible vertical lines have been removed from the tables
6. Table titles appear above the table

I trust that these revisions are all in order.

Sincerely,

Rhonda Orr